Author’s response to reviews

Title: Evaluation of a new high-dimensional miRNA profiling platform

Authors:

Julie M Cunningham (cunningham.julie@mayo.edu)
Ann L Oberg (oberg.ann@mayo.edu)
Pedro M Borralho (borralho@ff.ul.pt)
Betsy T Kren (krenx001@umn.edu)
Amy J French (french.amy@mayo.edu)
Liang Wang (wang.liang@mayo.edu)
Brian M Bot (bot.brian@mayo.edu)
Bruce W Morlan (morlan.brace@mayo.edu)
Kevin AT Silverstein (silve023@umn.edu)
Rod Staggs (stag004@umn.edu)
Yan Zeng (zengx003@umn.edu)
Anne-Francoise Lamblin (lambl001@umn.edu)
Christopher A Hilker (hilker.christopher@mayo.edu)
Jian-Bing Fan (jfan@illumina.com)
Clifford J Steer (steer001@umn.edu)
Stephen N Thibodeau (sthibodeau@mayo.edu)

Version: 8 Date: 14 August 2009

Author’s response to reviews: see over
August 14, 2009

Scott Edmunds PhD
Senior Editor
BioMed Central Editorial Production Team

Re manuscript ID: 980184624734351:

We have proof read manuscript and are submitting the revised version..

Sincerely,

Julie M. Cunningham, Ph.D.
Director, Genotyping Shared Resource
Co-Director, Gene Analysis Shared Resource
Advanced Genomic Technology Center
Mayo Clinic College of Medicine
Assistant Professor of Laboratory Medicine
Phone: 507 538 6863  Fax 507 266 0340
email: cunningham.julie@mayo.edu